Author’s response to reviews

Title: Maternal Vegetable Intake During and After Pregnancy

Authors:
Alison Tovar (alison_tovar@uri.edu)
Jill Kaar (JILL.KAAR@ucdenver.edu)
Karen Mccurdy (Kmccurdy@uri.edu)
Alison Field (alison_field@brown.edu)
Dana Dabelea (Dana.Dabelea@ucdenver.edu)
Maya Vadiveloo (maya_vadiveloo@uri.edu)

Version: 3 Date: 31 May 2019

Author’s response to reviews:

We have added the request information to the Ethics section as well as the Availability of data and clarified that an email is required in order to receive the data.